
Status of BigCAL
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SIX APPROVED EXPERIMENTS INTEND TO USE THE

CALORIMETER:

E04-108 (GEP-III), E04-019 (GEP-2G), E03-109 (SANE),

E04-113 (Semi-SANE),

E05-101 (helicity correlations in WACS), and

E03-003/PR06-001 (polarization transfer in WACS)� Status as of May 2005� Improving the detector during the summer� Current status

Andrew Puckett, GEANT simulation of BigCAL (background

rates and trigger issues)



May 2005 Status: tests with cosmics



May 2005 Status



May 2005 Status

0
25
50
75

100
125
150
175
200

50 100 150 200

ID
Entries

           1616
            959

  79.58    /    17
Constant   196.5
Mean   30.62
Sigma   8.861

ADC chan., x=16 y=16

0

25

50

75

100

125

150

175

200

50 100 150 200

ID
Entries

           3320
            995

  124.6    /    22
Constant   174.9
Mean   30.43
Sigma   9.932

ADC chan., x=33 y=20

0

20

40

60

80

100

120

140

50 100 150 200

ID
Entries

           3016
            925

  139.7    /    27
Constant   114.0
Mean   39.42
Sigma   13.68

ADC chan., x=30 y=16

0

20

40

60

80

100

120

140

50 100 150 200

ID
Entries

           3310
            644

  51.05    /    18
Constant   137.9
Mean   30.45
Sigma   8.506

ADC chan., x=33 y=10



May 2005 Status: Summary

10

20

30

40

50

10 20 30
0

5

10

15

20

25

30

35

40

peak position, ADC channels
x block

y 
bl

oc
k

0

100

200

300

400

500

600

0 20 40 60 80 100

ID
Entries
Mean
RMS

             22
           1743

  30.10
  2.136

peak position, ADC channels

10

20

30

40

50

10 20 30
0

5

10

15

20

25

30

35

40

sigma, ADC channels
x block

y 
bl

oc
k

0

100

200

300

400

500

600

0 10 20 30 40 50

ID
Entries
Mean
RMS

             23
           1743

  9.763
  1.981

sigma, ADC channels



Detector improvements: optical problems

Optical grease problems:� yellow color in few weeks exposed to air

– Don’t use ”Dow Corning Q2-3067” optical grease!

� difficult to clean

Using cookies (Hamlet Mkrtchyan and Razmik Asaturyan) was a

good solution



Detector improvements: mechanical problems
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� re-stacking of the glass blocks (Yuri Melnik, Yuri Gontcharenko)� drilling holes in the front plate (Arthur Mkrtchyan, Andrew Puckett)





Current Status: Summary
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Current Status

pedestal widths
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Current Status

about 15 % signal width deterioration
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Current Status

30 Volts - about 20 % signal change
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Current Status
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BigCAL STATUS C O N C L U S I O N S� The improvements will help the many experiments that will

use the calorimeter for:

– easier installation and maintenance

– faster UV curing procedure

– allows using monitoring system� Using cookies instead of optical grease deteriorates the

signal width by � 15 %� Current status:

– 1384 chan. ready (refurbished); remaining 360 to be

completed and tested by February 2006

– 912 signal cables and 720 HV channels are used in the

moment by GeN (HallA)� Plans for 2006:

– building and testing the trigger

– connecting the complete set of cables

– building the absorber (10cm Al)

– building the monitoring system

– software: off-line, data base, GEANT simulations



MANY THANKS TO

– Mark Jones, Charles F. Perdrisat, Vina Punjabi, Andrew

Puckett, Donald Goggans, Ryan Ford

– Yuri Melnik, Yuri Gontcharenko (re-stacking all the glass

blocks and re-installing the Protvino part)

– Hamlet Mkrtchyan, Razmik Asaturyan - backing cookis

(optical)

– Yuri Matulenko (off-line: automatic calibration with cosmics

and beam)

– Arthur Mkrtchyan (data base)


